



# REAL EXCHANGE RATE AND EXTERNAL PUBLIC DEBT IN EMERGING AND DEVELOPING COUNTRIES

Documents de travail GREDEG GREDEG Working Papers Series

**Grégory Donnat** 

#### **GREDEG WP No. 2025-15**

https://ideas.repec.org/s/gre/wpaper.html

Les opinions exprimées dans la série des **Documents de travail GREDEG** sont celles des auteurs et ne reflèlent pas nécessairement celles de l'institution. Les documents n'ont pas été soumis à un rapport formel et sont donc inclus dans cette série pour obtenir des commentaires et encourager la discussion. Les droits sur les documents appartiennent aux auteurs.

The views expressed in the **GREDEG Working Paper Series** are those of the author(s) and do not necessarily reflect those of the institution. The Working Papers have not undergone formal review and approval. Such papers are included in this series to elicit feedback and to encourage debate. Copyright belongs to the author(s).

## Real Exchange Rate and External Public Debt in Emerging and Developing Countries

Grégory Donnat\*

GREDEG Working Paper No. 2025–15

#### Abstract

Developing countries implement policies to maintain a stable exchange rate, with export-led growth strategies to spur economic growth and stability. The real exchange rate is a key factor of external competitiveness but can also cause economic and financial disruptions. These countries are dependent on external financing which impacts the real exchange rate movements over the medium and long run. We empirically explore the response of the real exchange rate to external public indebtedness in developing countries, from 1975 to 2017, using the iterative Bayesian shrinkage procedure to handle the cross-country differences in panel data. The contribution to the literature is twofold. First, we find that the change in the real exchange rate depends on the external public indebtedness in an inverted U-shape relationship in developing countries. Second, we determine an external debt threshold that minimizes changes in the real exchange rate for each country.

**Keywords:** Real Exchange Rate, External Debt, Developing Countries, Heterogeneity

**JEL Codes:** F31; F34; O16

<sup>\*</sup>Université Côte d'Azur, CNRS, GREDEG, France. mail: gregory.donnat@univ-cotedazur.fr

## 1 Introduction

Since the 1970s, the external competitiveness and promotion of manufactured exports have been crucial factors of economic growth in developing countries. East Asian countries (South Korea, Singapore, Taiwan) experienced considerable industrialization, expansion of their manufacturing sector and export diversification. Most emerging countries, which increased their manufactured exports, implemented export-led growth strategies with a consistent exchange rate policy. The main objective of an exportpromoting exchange rate policy is to limit the misalignments<sup>1</sup> of real exchange rate (EDWARDS, 1989; COTTANI et al., 1990; GHURA and GRENNES, 1993; RAZIN and COLLINS, 1997; RODRIK, 2008, RER).<sup>2</sup>

The RER is a fundamental relative price in developing economies that concerns a wide range of non-tradable goods.<sup>3</sup> Production and investment choices are also influenced by RER shifts. A stable RER helps to promote manufactured exports. In addition, the RER is more appropriate than the nominal exchange rate to assess the change of the exchange rate over the medium or long run. Examining the RER requires considering the international macroeconomic context. For example, the Behavioral Equilibrium Exchange Rate (CLARK and MACDONALD, 1999, BEER) or the Natural Real Exchange Rate (LIM and STEIN, 1998; STEIN, 1998, NATREX) highlight key macroeconomic factors that influence the equilibrium RER over the medium or long run. The literature shows that the RER is determined by the interaction between

<sup>&</sup>lt;sup>1</sup>The current real exchange rate persistently diverges from its long-run equilibrium value. The equilibrium real exchange rate is determined over a medium- or long-term horizon based on purchasing power parity approach, or a set of macroeconomic variables for example (see Section 2).

 $<sup>^{2}</sup>$  RODRIK (2008) shows that an undervaluation of the currency promotes economic growth. RAZIN and COLLINS (1997) find that persistent misalignments of RER are associated with lower long-term growth rate.

 $<sup>{}^{3}\</sup>text{RER}$  may also be defined as the domestic relative price of tradable to non-tradable goods, more appropriate for developing countries which are price-takers for tradable goods

trade factors and financial fundamentals, such as the country's net external position involving savings and public debt. In these models, the RER adjusts to ensure the internal and external macroeconomic equilibrium. External equilibrium refers to the situation in which there is a current account balance or an imbalance that is financed by a sustainable rate of capital flows (ARTIS and TAYLOR, 1993; FARUQEE, 1995). Theoretically, the sustainability of the capital account depends on the "desired" stocks of external assets and liabilities between countries (FARUQEE, 1995).<sup>4</sup> During the 1980's, the economic integration of developing countries, especially low- (LICs) and lower middle-income countries (LMICs), into global markets fostered private capital inflows.<sup>5</sup> For example, private capital inflows increased from USD 50 billion in 1987 to USD 150 billion per year in 1997 in developing countries (LARTEY, 2008). The external indebtedness of developing countries has also significantly increased, as reflected in the debt crises of developing countries in 1982 and the Heavily Indebted Poor Countries initiative in 1996.<sup>6</sup>

The purpose of this paper is to explore the relationship between the external public indebtedness and the RER over the long run in developing countries. There is no common agreement in the theoretical and empirical literature about this relationship. On the one hand, some studies point out that a country in need of external financing experiences a real currency depreciation to reduce its current account deficit by generating a trade surplus. The real exchange rate depreciation is required to stabilize debt-to-GDP ratio.<sup>7</sup> On the other hand, empirical works show that external public

<sup>&</sup>lt;sup>4</sup>A desired stock of external assets and liabilities represents the ideal level that a country aims to reach in terms of investments and debt to ensure external economic stability, while avoiding imbalances that could lead to balance of payments or debt crises.

<sup>&</sup>lt;sup>5</sup>The income classification is defined according to the World Bank.

<sup>&</sup>lt;sup>6</sup>The HIPC initiatives launched by the IMF, World Bank and Paris Club have resulted in a significant amount of debt cancellation for low-income countries that were considered over-indebted at the time.

<sup>&</sup>lt;sup>7</sup>A real currency depreciation improves the current account deficit, which in turn reduces external

debt is associated with a real currency appreciation due to a bias of public spending towards non-tradable goods (SEKKAT and VAROUDAKIS, 2000). More recent studies show that the relationship between the public debt and the RER depend on the debt sustainability. The debt sustainability is assessed through the external public debt ratio relative to GDP in this study. The debt dynamics, or cointegration estimates between debt series and macroeconomic fundamentals are alternative approaches to assess debt sustainability.

We investigate the existence of a non-linear relationship between the external public debt and RER over the long run in developing countries. Only a few papers in the empirical literature assess this relationship. Most existing studies focus on developed rather than developing countries. We consider low- and lower middle-income countries, because they are mainly dependent on external financing. This study covers the time period from 1975 to 2017 for 34 developing countries. In order to deal with heterogeneity and handle the cross-country differences between developing countries, we estimate the relationships with the Pooled Mean Group estimator. Then, we apply the iterative Bayesian shrinkage procedure to estimate the relationship between the external public debt and RER country by country.<sup>8</sup> To our knowledge, this has never been done before in the literature.

We contribute new empirical findings to the literature. We find an inverted U-shape relationship between the external public debt and RER. A limited external public debtto-GDP ratio induces an appreciation of the RER through the demand for non-tradable goods, whereas a high external debt ratio leads to a real currency depreciation. The depreciation of the RER is expected to restore external macroeconomic balance. Since the inverted U-shape relationship between the external public debt and RER is estimated country by country, we can determine a turning point for each country. In other

financing needs.

<sup>&</sup>lt;sup>8</sup>Specifically, we consider the real effective exchange rate in the study.

words, we identify a public debt threshold at which the relationship between external public debt and RER shifts from positive to negative. We thus propose an alternative approach to define a *stabilizing* external public debt. This level of external public debt would minimize changes in the RER.

The rest of this paper is organized as follows. Section 2 discusses the literature on determinants of real exchange rate and external debt. Section 3 explains the data and issues related to estimating the real exchange rate. Section 4 presents the empirical strategy. Section 5 analyzes the results of the estimates with the Pooled Mean Group. Section 6 demonstrates the relationship between RER and external public debt country by country. Section 7 describes the robustness checks conducted to strengthen the study. Section 8 concludes.

## 2 Related Literature

There are several approaches to the determination of equilibrium exchange rate (FRENKEL and GOLDSTEIN, 1986):

- i. the monetary models and portfolio balance models,
- ii. the purchasing power parity (PPP) approach where nominal exchange rate change is needed to offset price level ratio changes in order to maintain a constant RER.

For FARUQEE (1995, p.81): "it may be quite misleading to view the real exchange rate as an isolated measure of external competitiveness without further reference to developments within the overall macroeconomic environment". This remark fits with the third approach of RER determination:

iii. the "underlying balance models". In these models, the equilibrium RER is not

steady and changes as a function of fundamentals. It adjusts to achieve internal and external macroeconomic balances over the medium or long run.

On the one hand, the equilibrium RER is the rate that would satisfy the desired internal and external macroeconomic balance. This is a normative approach to the equilibrium RER (WILLIAMSON, 1983; WILLIAMSON, 1994, Fundamental Equilibrium Exchange Rate).<sup>9</sup> On the other hand, the equilibrium RER can be defined with a positive approach where the RER changes depending on macroeconomic fundamentals, such as terms of trade, trade openness, productivity, and net foreign asset position, resulting in the exchange rate that is needed and relevant in the medium and long run. The Behavioral Equilibrium Exchange Rate<sup>10</sup> (CLARK and MACDONALD, 1999, BEER) and the Natural Real Exchange Rate<sup>11</sup> (LIM and STEIN, 1998, NATREX) correspond to this approach. Internal balance corresponds to the situation when national economic output is at its potential level. External balance may refer to various situations. If the current account balance is positive, the country is a net lender or has a capacity of financing other countries, otherwise the country is a net borrower, with capital inflows. Therefore, over the medium run, the macroeconomic equilibrium is characterized by an internal balance and the condition of external balance where a current deficit is financed by sustainable capital inflows (excluding speculative capital flows). If the country's current account is in deficit, the RER becomes unsustainable in the long run because of an accumulation of external debt. A real currency depreciation, through a drop in the country's net wealth and consumption, allows for a trade surplus

<sup>&</sup>lt;sup>9</sup>The FEER approach relies on significant normative conditions that are subject to discussions and hard to represent empirically, such as the determination of current account targets (considered as desired or sustainable).

<sup>&</sup>lt;sup>10</sup>The BEER is easy to implement but its lack of underlying theoretical mechanisms has been called into question.

<sup>&</sup>lt;sup>11</sup>The NATREX is based on robust equilibrium foundations. It allows us to estimate short-, mediumand long-term equilibrium exchange rates dynamically. The NATREX is a dynamic approach of FEER, in which the current account target is modeled based on its fundamental determinants.

to service the debt.

We will use the BEER approach, especially relevant for developing countries, to estimate the relationships between RER and macroeconomic fundamentals, such as external public debt over the medium and long run. In this approach, the country's external position is a key determinant of the real exchange rate in the medium and long term. Developing countries are characterized by negative net external positions that result in external public indebtedness. The countries' total public debt is divided into internal and external debt according to the type of creditors or the denominated currency. We consider the external public debt because financial markets are less deep in developing countries. Moreover, the debt is denominated in foreign currency (often in dollars or euros). External public financing is an economic opportunity for governments to increase investment, consumption and economic growth, but the debt burden is also a pressure on national income and budget (BACHA, 1990), therefore the sustainability of this indebtedness is an important issue. The debt sustainability can be assessed in several ways:

- i. the debt burden is measured by calculating ratios between debt indicators and resource indicators (GDP, exports of goods and services, foreign exchange reserves). The public debt is considered as unsustainable if it exceeds a given threshold. For example, this ratio is set at 60% of GDP in the European Union. In the past, the IMF and the World Bank have considered situations of over-indebtedness (or debt distress) by looking at the present value of external debt in ratio to exports or revenues for example, especially for the Heavily Indebted Poor Countries initiative.
- ii. An analysis of debt dynamics (a forward-looking approach) that relies on the pri-

mary deficit and the difference between the interest rate and the economic growth rate. According to these two elements, we can assess if the debt is sustainable or if it follows an explosive trajectory due to a snowball effect.

iii. A debt sustainability analysis using a cointegration relationship between a debt indicator and a set of macroeconomic fundamentals.

The literature, both theoretical and empirical, is not conclusive on the relationship between the RER and external public debt over the long run. Moreover, the literature highlights opposite effects induced by external indebtedness.

On the one hand, empirical works show a negative relationship between external public debt and RER in extension of the BEER approach and the Edwards' model (EDWARDS, 1988). MONGARDINI (1998) highlight that a current account deterioration due to an increasing debt-service ratio led to a real currency depreciation in Egypt from 1987 to 1996. Conversely, a reduction in the debt level increased the wealth in each period and the demand for tradable and non-tradable goods, which raised price level of non-tradables. Thus, debt reductions (from debt relief for example) led to a real currency appreciation. Moreover, COUDERT (1999) produced estimates for a panel of emerging countries with the OLS method and showed that an increase in debt ratio of 1 point of GDP led to a depreciation of about 0.3-0.5%. It is also possible to assess the effects of debt policy on real exchange rate thanks to the New Open Economy Macroeconomics (NEOM) framework. The model developed by GANELLI (2005) as an extension of the Overlapping Generations (BLANCHARD, 1985) and Redux models (OBSTFELD and ROGOFF, 1995), showed that a debt-financed fiscal policy results in a real currency depreciation over the long term, due to a decline of net foreign assets.

On the other hand, some studies highlight a positive relationship between external financing and real exchange rate (YANO and NUGENT, 1999; SEKKAT and VAROUDAKIS, 2000). Governments in developing countries that have access to capital markets use external funds to increase social expenditures (wealth, education or infrastructures). This generates an excess of demand in non-tradable goods, at the expense of tradable goods, that leads to a rise in relative price of non-tradable goods and a real appreciation. Empirically, SEKKAT and VAROUDAKIS (2000) show that a sustainable indebtedness results in an appreciation of the equilibrium real exchange rate over the long term.

Finally, some studies show that there is a non-linear relationship between the external public debt and RER that depends on the debt sustainability. Using a model based on the current account dynamics (BUITER, 1988; FRENKEL and RAZIN, 1996), SENE and WANE (2013) show that a sustainable debt<sup>12</sup> is associated with an appreciation of the RER to ensure the adjustment towards equilibrium, while an unsustainable debt leads to a depreciation of the RER in the long run.<sup>13</sup> In addition, the NATREX model may also explain an inverted U-shape relationship between external public debt and RER. An internal shock causes external indebtedness (and a deterioration of the capital account) leading to a real currency appreciation. This currency appreciation reduces the investment ratio and thus the capital stock ratio. The consumption starts to decline when the external public debt reaches a critical high level.<sup>14</sup> The fall in consumption reduces the trade deficit which improves the current account. However, if these adjustment effects are weak, then a depreciation of the RER is required to improve the current account in the long run and stabilize the external debt ratio (COUHARDE et al., 2016).

The purpose of this paper is to investigate a potential link between the external

<sup>&</sup>lt;sup>12</sup>The public debt is considered as unsustainable if the real interest of commitments is higher than the GDP growth rate.

<sup>&</sup>lt;sup>13</sup>See FRENKEL and RAZIN (1996) and SENE and WANE (2013) which describe the current account dynamics model and the long-run adjustment of the real exchange rate depending on the debt sustainability.

<sup>&</sup>lt;sup>14</sup>The external indebtedness is accompanied by wealth de-accumulation.

public debt and the RER, in particular to determine if there exists an inverted U-shape relationship (see Figure 1).<sup>15</sup> In first instance, we highlight long-run relationships between the RER and macroeconomic fundamentals, including external debt, with the Pooled Mean Group estimator. If such a relationship exists, then the derivation of a turning point will allow us to define a *stabilizing* external public debt for each country, thanks to the Bayesian iterative procedure which allows us to get the country-bycountry relationships. In other words, we derive a level of external public debt that minimizes changes in the RER. We assume that the turning point (i.e., the threshold of external indebtedness at which the real exchange rate shifts from an appreciation to a depreciation) varies between countries. In the following section, we present the data, and especially how the real exchange rate is defined.

Figure 1: An inverted U-shape relationship between RER and External Public Debt Real Exchange Rate



<sup>&</sup>lt;sup>15</sup>The purpose of this figure is to point out a dynamic relationship between the real exchange rate and the external public debt. Depending on the external public debt level, the real exchange rate depreciates or appreciates. However, we do not pretend to show an effect of the real exchange rate on the external public debt, nor that the real exchange rate values depend on the external public debt level (for example the same real exchange rate value for two distinct levels of external public debt).

## **3** Data and preliminary analysis

#### 3.1 Data

We collected the outstanding external public and publicly guaranteed debt and debt service on total external public debt from the World Bank International Debt Statistics, and the Gross Domestic Product (in current USD) from World Development Indicators. In this paper, the external public debt and debt service to GDP ratios measure the indebtedness of developing countries and are included in the set of explanatory variables used to explore the effects of external public debt on real effective exchange rate (REER). We also collect from the IMF and World Bank databaset, GDP per capita relative to the United States as a proxy for productivity (RODRIK, 2008), the countries' trade openness (trade flows relative to GDP), the net foreign asset position relative to GDP, and the terms of trade.

The real effective exchange rate is the rate at which a country's currency trades against a basket of other currencies in real terms. We retrieved data on REER from the CEPII database "EQCHANGE" (COUHARDE et al., 2018). The database provides wide information (country and time coverage) on: (i) effective exchange rates (NEER and REER); (ii) estimates of equilibrium RER and currency misalignments. "The REER of country i in period t is calculated as the weighted average of real bilateral exchange rate against each of its N trading partners" (COUHARDE et al., 2018, p.9):

$$REER_{i,t} = \prod_{j=1}^{186} \left( \frac{NER_{i,t} \times P_{i,t}}{P_{j,t}} \right)_{ij,t}^{w_{i,t}}$$
(1)

where  $\frac{NER_{i,t} \times P_{i,t}}{P_{j,t}}$  is an index of real exchange rate (base 100 = 2010) of the currency of country *i* vis-à-vis the currency of its trading partner *j* in period *t*.  $P_{i,t}$  and  $P_{j,t}$ 

are the price index of countries i and j, while  $w_{ij,t}$  is the weight of the j-th partner in the bilateral trade of country i. An increase (decrease) in REER corresponds to a real currency appreciation (depreciation). We selected the broad index with 186 partners and the fixed weights system from 1973.<sup>16</sup> The authors show that their effective exchange rates are robust and consistent with those provided by the IMF, World Bank, or Bruegel institution. Moreover, the database provides more information on REER for developing countries, especially LICs, and a larger time period coverage than the IMF's International Financial Statistics.

#### 3.2 Country panel and heterogeneity

All data are annual, cover the period from 1975 to 2017 and extend to 37 developing countries (Table A.1 in the Appendix): Bangladesh (BGD), Benin (BEN), Botswana (BWA), Burkina Faso (BFA), Burundi (BDI), Cameroon (CMR), Central African Republic (CAF), Chad (TCD), Republic of Congo (COG), Cote d'Ivoire (CIV), Dominican Republic (DOM), Egypt (EGY), El Salvador (SLV), Gabon (GAB), The Gambia (GMB), Guatemala (GTM), Guyana (GUY), Honduras (HND), Jamaica (JAM), Kenya (KEN), Lesotho (LSO), Madagascar (MDG), Mali (MLI), Mauritius (MUS), Mauritania (MRT), Nepal (NPL), Niger (NER), Pakistan (PAK), Panama (PAN), Paraguay (PRY), Philippines (PHL), Rwanda (RWA), Senegal (SEN), Sri Lanka (LKA), Thailand (THA), Togo (TGO) and Tunisia (TUN).

These Low- and Lower- Middle-Income Countries,<sup>17</sup> according to the World Bank classification, are heterogeneous in terms of country size, geographical position, economic performance or trade integration, and commodities endowment.<sup>18</sup> Moreover,

 $<sup>^{16}</sup>$ See COUHARDE et al. (2018) for further methodological features about weighting scheme, narrow indexes, etc.

 $<sup>^{17}{\</sup>rm We}$  build a sub-sample of countries that are classified as LICS and LMICs throughout the time period (see Table A.1 in the Appendix).

<sup>&</sup>lt;sup>18</sup>The World Bank classifies countries into four groups depending on the GNI per capita (in USD):

some countries are members of economic and monetary regional communities and unions. On the one hand, this regional integration leads to different levels of financial development or trade liberalization, and changes the effects of international capital flows, external position or public debt on RER.<sup>19</sup> On the other hand, these countries implement different exchange rate regimes. Members of the West African Economic and Monetary Union have kept their currency pegged to the French Franc and Euro, while the West African English-speaking countries have moved toward independently floating currencies. Therefore, the implications of external public debt on the REER may depend on the exchange rate arrangement implemented. RER adjustments are slower in fixed exchange rate regimes. However, a fixed exchange rate can enhance the appreciation effects of capital inflows on real exchange rates (COMBES et al., 2012). Table A.2 in the Appendix describes the exchange rate regime arrangements implemented over the period. In addition, some countries in the panel have benefitted from debt reductions through the HIPC and MDRI initiatives (see Table A.3 in the Appendix). Heavily Indebted Poor Countries (HIPCs) have been granted debt reductions in order to recover sustainable debt levels.

#### 3.3 Preliminary analysis

In order to explore the effects of external debt on real exchange rate, we collect available data on equilibrium real exchange rates from CEPII's "EQCHANGE" database. We seek to observe any correlations between external public debt-to-GDP ratio and equilibrium real exchange rate, over the medium or long term. We retrieve from the CEPII database the computed equilibrium exchange rates that are based on the BEER approach (CLARK and MACDONALD, 1999) which assesses long-run determinants (funda-

Low income, Lower middle income, Upper middle income, High income.

<sup>&</sup>lt;sup>19</sup> ROY and DIXON (2016) show that trade openness mitigates the appreciation effect of remittances on real exchange rates. ACOSTA et al. (2009) find similar results with deeper financial systems.

mentals) of equilibrium real exchange rate.<sup>20</sup> EDWARDS (1988) and ELBADAWI (1994) developed suitable models to explore the fundamentals underlying long-run movements of the RER of developing countries, such as the terms of trade, government consumption, trade restrictions, technological progress or external debt. Figure A.1 in the Appendix connects the changes of external public debt-to-GDP ratio and equilibrium real exchange rate (ERER) from 1975 to 2017 for 10 countries in the sample. We see, for Benin, that substantial external indebtedness is coupled with a decrease in the ERER from 1975 to the end of 1990s, whereas a low or limited external debt ratio is linked with an appreciation of ERER. For Burundi, an ERER depreciation is associated with a sharp increase of debt to GDP ratio. We observe the same path for other HIPCs (Cameroon, Central African Republic, Guyana, Madagascar, Mali) and non-HIPCs. Moreover, a decrease of external public debt level (due to debt relief or not) is related to an appreciation of ERER for all countries (except Kenya). Finally, the annual variation in the real exchange rate is connected to the level of external public debt for the full period and the sample of countries studied in Figure A.2 in the Appendix. We observe a negative relationship between the level of external public debt and variations in the real exchange rate. Changes in the real exchange rate become negative above a debt level of around 20% of GDP. The following section describes the empirical strategy.

<sup>&</sup>lt;sup>20</sup>The authors include in their regressions as fundamentals of the RER: Balassa-Samuelson effect, the economy's net foreign asset position and the economy's terms of trade.

# 4 Real Exchange Rate and Economic Fundamentals: long-run relationship with Pooled Mean Group

As we have seen before, the panel of developing countries is characterized by heterogeneity, both between countries and over time. There may also be heterogeneity in changes in the real exchange rate or the external public debt of these countries. We assume that the macroeconomic fundamentals converge as countries develop economically, and their variations tend to be similar between countries over the long term. The pooled mean group (PMG) estimator, introduced by PESARAN et al. (1999), allows this assumption and econometric specification. This estimator allows that the intercepts and coefficients differ between countries over the short run, but the coefficients are constrained to be identical over the long run, unlike the mean group (MG) estimator that does not consider potential homogeneity across countries (PESARAN and SMITH, 1995). As we discussed in Section 2, the real effective exchange rate can be explained in the long run by a set of macroeconomic fundamentals. In this section, we explore the long run relationship between the REER and the external public debt in addition to a set of fundamental macroeconomics with the following equation:

$$\operatorname{REER}_{i,t} = \mu_i + \beta_i Z_{i,t} + \delta_t + \varepsilon_{i,t} \tag{2}$$

where  $REER_{i,t}$  is the real exchange rate in logarithm for country *i* in period *t*, considering 186 trade partners and fixed weights for 1973-2017, as detailed in the previous section.  $Z_{i,t}$  is a set of macroeconomic fundamentals and the external public debt-to-GDP ratio for each country *i* and period *t*,  $\mu_i$  is the constant,  $\delta_t$  corresponds to time fixed effects and  $\varepsilon_{i,t}$  is the error term. We employ an unrestricted error-correction autoregressive distributed lag (ARDL) model. The ARDL (1;1) can be written as the following equation:

$$\Delta REER_{i,t} = \phi_i (\text{REER}_{i,t-1} - \theta_0 - \theta_1 \text{TOT}_{i,t} - \theta_2 \text{Trade}_{i,t} - \theta_3 \text{Prod}_{i,t} - \theta_4 \text{NFA}_{i,t} - \theta_5 \text{Debt}_{i,t}) - \delta_{1i} \Delta \text{TOT}_{i,t} - \delta_{2i} \Delta \text{Trade}_{i,t}$$
(3)  
$$- \delta_{3i} \Delta \text{Prod}_{i,t} - \delta_{4i} \Delta \text{NFA}_{i,t} - \delta_{5i} \Delta \text{Debt}_{i,t} + \varepsilon_{i,t}$$

The PMG estimator imposes that  $\theta_i$  is the same for all countries. The terms in parentheses are the long run coefficients.  $\phi_i$  is the coefficient of adjustment to long run equilibrium given by  $\phi_i = -(1 - \lambda_i)$ . The set of fundamental macroeconomics parameters is composed of the terms of trade  $(TOT_{i,t})$ , trade openness  $(Trade_{i,t})$ , productivity  $(\operatorname{Prod}_{i,t})$ , and net foreign asset position  $(\operatorname{NFA}_{i,t})$ . A rise in the terms of trade can lead to ambiguous effects: if the income effect dominates the substitution effect, such a rise increases the national domestic income and the demand for non-tradable goods which induces a real appreciation; otherwise (i.e., if the substitution effect dominates) the increase in national domestic income boosts demand for imported goods at the expense of non-tradable goods and generates a real currency depreciation. These two effects may cancel each other out and explain the absence of a terms of trade impact in some countries (LARTEY et al., 2012). There is also an ambiguous effect of trade restrictions on the real exchange rate. The income effect has a negative influence on the REER: a rise in the imported goods price affects national income and reduces demand for tradable and non-tradable goods, leading to a real depreciation. If the substitution effect dominates, imported goods are substituted by non-tradable domestic goods inducing an appreciation of the equilibrium REER. It is commonly agreed that the income effect dominates the substitution effect (EDWARDS, 1988), hence restricting trade induces a fall of the price ratio of tradable to non-tradable goods, leading to a real appreciation.

Productivity attempts to capture the Balassa-Samuelson effect, which is proxied by the GDP per capita relative to the United States (ROGOFF, 1996; RODRIK, 2008). The Balassa-Samuelson effect is positive on the REER as a country is economically growing. In particular, the tradable sectors grow faster than the non-tradable sectors which induces higher wages in tradable sectors, which in turn puts upward pressure on wages in non-tradable sectors. This rise of wages in non-tradable sectors raises the relative price for non-tradable sectors, because prices in tradable sectors are determined by international markets, and leads to a real appreciation. Finally, an increase in the net foreign asset position of the country has a positive influence on the REER.<sup>21</sup>

Including stock variables in the analysis implies that flow equilibrium must follow as a necessary condition.<sup>22</sup> In our model, we consider factors that impact the country's trade position on international markets and the propensity of the country to be net lender or borrower. FARUQEE (1995) highlights that the interaction between structural components of both the current account and the capital account jointly determine the RER over the long run. Hence, Debt<sub>*i*,*t*</sub> corresponds to the logarithm of the external public debt-to-GDP ratio. We include in Equation 4 the squared variable of Debt<sub>*i*,*t*</sub> as a proxy for high external public debt levels in order to explore if the movements of the real effective exchange rate vary depending on the external debt-to-GDP ratio. The relevance of using a debt ratio is justified since we are looking for a statistical relationship between the debt ratio and the evolution of the RER.

 $<sup>^{21}</sup>$  LANE and MILESI-FERRETTI (2000) show that the net foreign asset position is an important determinant of the real exchange rate in both developing and developed countries.

 $<sup>^{22}</sup>$  FARUQEE (1995) illustrates this idea with the transfer problem: a fall in the NFA stock for a country leads to a decline in domestic spending and demand for domestic goods due to the redistribution of wealth. The international transfer of wealth must be accompanied by a real depreciation to improve the trade position in order to offset losses in interest revenues from abroad (external equilibrium).

$$\Delta REER_{i,t} = \phi_i (\text{REER}_{i,t-1} - \theta_0 - \theta_1 \text{TOT}_{i,t} - \theta_2 \text{Trade}_{i,t} - \theta_3 \text{Prod}_{i,t} - \theta_4 \text{NFA}_{i,t} - \theta_5 \text{Debt}_{i,t} - \theta_6 \text{Debt}_{i,t}^2) - \delta_{1i} \Delta \text{TOT}_{i,t} - \delta_{2i} \Delta \text{Trade}_{i,t} - \delta_{3i} \Delta \text{Prod}_{i,t} - \delta_{4i} \Delta \text{NFA}_{i,t} - \delta_{5i} \Delta \text{Debt}_{i,t} - \delta_{6i} \Delta \text{Debt}_{i,t}^2 + \varepsilon_{i,t}$$

$$(4)$$

Therefore, this model allows us to explore a non-linear relationship between the RER and the external public debt in the developing countries. The next section presents in detail the results of the estimates.

### 5 Results

Table A.4 in the Appendix displays unit root tests (Augmented Dickey-Fuller and IPS) on the REER, external public debt and other macroeconomic aggregates.<sup>23</sup> Unit root tests confirm that some variables are nonstationary and can be considered as integrated of order one. We assume the potential presence of cross-sectional dependence in the sample. Consequently, we also test the stationarity of the series using second-generation tests (PESARAN, 2003). Table 1 presents the estimates of the long-run relationship between REER and macroeconomic fundamentals.<sup>24</sup> Coefficients in columns 1-2 and 4-5 are estimated using the Pooled Mean Group method, while coefficients in columns 3 and 6 are estimated using a dynamic common-correlated effects model to control for the presence of cross-sectional dependence.

 $<sup>^{23}</sup>$ IPS refers to the IM et al. (2003) unit root test. IPS introduced heterogeneity : the IPS test is a way to combine the evidence from N unit root tests for N different individuals.

<sup>&</sup>lt;sup>24</sup>We note that the Hausman test confirms that the long-run parameters are homogeneous, hence the pooled mean group might be preferable to the mean group estimator and lead to more efficient estimates.

The error correction term is negative and statistically significant. This result is consistent with the existence of a cointegrating vector and excludes potential omitted variable bias. We note that, for the Low- and Low-Middle Income Countries panel, the speed of adjustment is statistically significant and higher than one for all countries.<sup>25</sup> Indeed, these countries are further from the equilibrium values relative to emerging countries. The relationship between the REER and its macroeconomic fundamentals is consistent with our expectations. Indeed, the Balassa-Samuelson effect, measured by the relative GDP per capita, is positive and statistically significant. The effect of the terms of trade on REER is also positive. The income effect tends to dominate the substitution effect. Regarding the implications of trade openness, we find that the estimated coefficient is not statistically significant. The lack of significance may support the idea that the income and substitution effects can cancel each other out. Finally, as expected, the NFA position has a positive and significant influence on the REER. In the same logic as NFA (i.e., a stock variable that measures the country's external position).

The results in Table 1 suggest a long-run, inverted U-shape relationship between external public debt and the REER for LICs and LMICs.We note that the estimated coefficients (in absolute value) associated with external public debt are higher for the LICs and LMICs panel. Relying on external funding allows for the financing of social expenditure (i.e., towards the non-tradable sector) which induce an appreciation of the REER. Conversely, a high external debt leads to a depreciation of the REER in order to ensure external balance and finance current account deficits.In order to ensure the sustainability of the external account, external public indebtedness is associated with a real currency depreciation to generate a trade surplus to service the debt.

<sup>&</sup>lt;sup>25</sup>This result for a smaller and more homogeneous sample of countries is consistent with the presence of a cointegrating vector, and thus that it is appropriate to impose long-run homogeneity.

	(1)	(2)	(3)	(4)	(5)	(6)
Dependent variable:	Log(Real I	Effective Exc	change Rate)			
	PMG		DCCE	PMG		DCCE
Panel:	All countri	ies		LICs and l	LMICs	
Error-correction term	-0.128***	-0.126***		-0.162***	-0.153***	
	(-8.80)	(-7.28)		(-8.52)	(-8.29)	
Log(Terms of Trade)	$3.677^{***}$	$2.939^{***}$	-1.723	0.242	$2.533^{**}$	1.649
	(4.35)	(2.81)	(-0.2)	(0.21)	(2.47)	(0.17)
Log(Trade)	0.019	-0.047	-0.233*	$0.160^{***}$	0.084	0.095
	(0.32)	(-0.96)	(-1.90)	(2.65)	(1.41)	(0.67)
Log(Productivity)	0.501***	$0.516^{***}$	0.014	0.035	0.210***	0.105
	(7.17)	(9.64)	(0.07)	(0.53)	(4.57)	(0.47)
NFA	0.008***			0.001		
	(5.17)			(1.62)		
External Public Debt	-0.134***	$0.283^{**}$	$1.147^{**}$	-0.071***	$0.419^{***}$	0.553
	(-3.81)	(2.47)	(2.11)	(-2.64)	(3.15)	(1.02)
External Public Debt <sup>2</sup>		-0.062***	-0.169**		-0.078***	-0.086
		(-3.46)	(2.15)		(-3.87)	(-1.23)
Cointegration test						
Panel rho	2.066	3.479				
Group rho	2.508	3.842				
Observations	1411	1426	1391	949	949	940
No of countries	34	34	34	23	23	23

Table 1: Long-run relationship between the REER, external public debt and macroeconomic fundamentals

Notes: Numbers in parenthesis are t-statistics. Short-run relationships are not reported in the table. We follow the PEDRONI (2004) cointegration tests. The null hypothesis assumes for each unit of the panel that the variables are not cointegrated. Pedroni's test allows for heterogeneous variances between countries in each period. The alternative hypothesis assumes for each unit of the panel that there is a cointegration vector, which can be different for each country. All specifications include a maximum of one lag. The total number of lags is chosen according to the Schwarz criterion (BIC). PMG and DCCE correspond to Pooled Mean Group and Dynamic Common Correlated Effects estimates respectively. \*, \*\* and \*\*\* denote significance at 10%, 5% and 1% levels. Low and Lower Middle-income countries panel is composed by the 26 developing countries presented in section 3, excluding Botswana, Guyana and Lesotho.

Therefore, Pooled Mean Group estimates provide evidence of the long-run relationship between the real exchange rate and macroeconomic fundamentals. However, the main limit of the Pooled Mean Group is that it provides a unique coefficient (a single slope) for the entire sample, whereas, the real exchange rate needs to be considered separately for each country. Therefore, in the next section, we apply bayesian shrinkage estimators which allow both to control for heterogeneity between countries and to get coefficients country by country.

# 6 Real Exchange Rate and External Public Debt: an inverted U-shape relationship

In this section, we apply the Empirical Iterative Bayes' estimator suggested by MAD-DALA et al. (1997), since it provides heterogeneous (country by country) estimates for the relationship between external public debt and real exchange rate. This long-term relationship is given by the following equation with a dynamic panel data model:

$$\Delta REER_{i,t} = c_i + \beta_i \operatorname{REER}_{i,t-1} + \delta_i \operatorname{Debt}_{i,t} + \theta_i \left(\operatorname{Debt}\right)_{i,t}^2 + \varepsilon_{i,t}$$
(5)

where  $\text{REER}_{i,t}$  is the real exchange rate (in logarithm) for country *i* in period *t*. Debt<sub>*i*,*t*</sub> corresponds to the external public debt to GDP ratio (in log).  $c_i$  and  $\varepsilon_{i,t}$  are the constant and the error terms, respectively. We include the squared variable of  $\text{Debt}_{i,t}$  as a proxy for high external public debt levels, in order to explore the potential inverted U-shape relationship between REER and external debt-to-GDP ratio. The inclusion in the model of the current variable (instead of the lagged variable) of external public debt to GDP, does not raise an issue, since in the dynamic panel data model framework, the empirical literature underlines that problems associated with time series are mainly related to the handling of heterogeneity. Moreover, the recent empirical literature (HSIAO et al., 1998; HSIAO, 2022) argues in favor of shrinkage estimators that shrink the heterogeneous estimators towards the pooled estimator.<sup>26</sup>

The models also include the lagged independent variable in the set of explanatory variables, in order to capture the convergence of REER towards its long-run equilibrium value and potential effects of economic fundamentals on REER in the long term, such as productivity, terms of trade, economy's net wealth, etc., which are not included in the model. These estimations explore the relationship between REER and external public debt for 26 developing countries (LICs and LMICs in Table A.1 in Appendix), from 1975 to 2017, while considering some issues related to real exchange rate estimates, as discussed in the following.

The Empirical Iterative Bayes' estimator presented in more detail in Appendix B allows us to deal with some issues associated with the real exchange rate estimates.

i. Considering the dynamics of exchange rates: The real exchange rate of a country is supposed to converge towards its equilibrium value over time. A way to incorporate these dynamics is to include the lagged independent variable into the set of explanatory variables (ROGOFF, 1996), although it can generate new sources of estimation bias and endogeneity. Using the Bayesian shrinkage estimator helps to resolve these problems. Indeed, for MADDALA et al. (1997), the empirical iterative Bayes' estimators are preferred if the model contains lagged endogenous variables (as it is the case in dynamic models). These estimators have more plausible values than the heterogeneous estimators. MADDALA and HU (1996) presented some Monte Carlo evidence to suggest that the iterative procedure gives

 $<sup>^{26}</sup>$ For instance, BALTAGI et al. (2008) consider the dynamic version of the classical Tobin investment model.

better estimates for panel data models.

- ii. Stationarity in heterogeneous panel data: non-stationary time series with traditional econometric techniques may generate spurious regression results (NEL-SON and PLOSSER, 1982). Empirical Bayes' estimators do not suffer from nonstationarity bias (HSIAO et al., 1998; BALTAGI et al., 2008).
- iii. Cross-country heterogeneity: in the previous section, we highlighted the crosscountry heterogeneity in the panel due to economic and demographic characteristics or membership of various economic and monetary communities. Moreover, the real exchange rate may be defined as the ratio between tradable and non-tradable prices, thus domestic relative prices are different between countries. COUDERT (1999) shows that indebtedness explains the changes of RER within a country more than differences between countries; thus, it legitimates using the empirical Bayes' estimator which provides country-by-country coefficients. These estimators allow to deal with heterogeneity and provide more stable, and less dispersed coefficients.

Tables B.1 and B.2 in the Appendix report the parameters obtained with the Bayesian shrinkage estimators for the model given by the Equation 5 for 1975-1999 and 1975-2017. For both estimations, we find some new empirical results. The coefficients  $\delta_i$  associated with the external public debt-to-GDP ratio are positive and statistically significant.

Indeed, an increasing external debt-to-GDP ratio is associated with an appreciation of the REER in Low- and Lower Middle-Income Countries, only as long as the debt level remains limited. In other words, a limited external public indebtedness is associated with a real currency appreciation. Developing countries' external financing results in capital inflows into the country. Firstly, these funds denominated in hard currency (US Dollars, Euros, Livres, Yen, etc.) are converted into domestic currency. Secondly, governments increase social expenditures. This spending increases the demand for non-tradable goods which raises the relative price of non-tradable to tradable goods and results in a real currency appreciation. The coefficients  $\theta_i$  associated with the squared variable of external debt ratio are negative and statistically significant for both estimations. A high external public indebtedness is associated with a real currency depreciation in LICs and LMICs over the medium and long run. This result is consistent with the long-run determinants of real exchange rate models (EDWARDS, 1989, FEER and NATREX) which highlight that indebtedness is not sustainable over the long term and lead to a depreciation of the real exchange rate in order to improve external balance. Moreover, a significant debt burden requires the government to reduce social expenditures to service the debt. The drop in demand for non-tradables lowers the relative price of non-tradable goods and leads to a real currency depreciation. Hence, there is an inverted U-shape relationship between the external public debt and RER in developing countries: a low external public indebtedness is associated with a real currency appreciation, whereas a high external public debt induces a real currency depreciation.

The NATREX model provides some insight to explain the inverted U-shape relationship between external public debt and REER. Indeed, the adjustment process of the real exchange rate from its medium-term to the long-term value is not linear and depends on the speed of adjustment of the stock variables such as net foreign assets, and especially the dynamics of wealth accumulation. However, if these adjustment effects are weak, then a depreciation of the RER is required to improve the current account in the long run (as mentioned Section 2). Having identified an inverted U-shape relationship between external public debt and the RER, it is relevant to determine a turning point for each country. This turning point corresponds to a threshold of external public debt to GDP ratio that minimizes the change in the RER.

## 7 Robustness checks: sensitivity to the country panel and external public debt threshold

In this section, we modify the panel of countries (see bottom panel in Table A.1 in the Appendix) in order to examine if we find similar relationships between the external public debt and the real exchange rate as in the previous section. Thus, we test the sensitivity of the results to the sample of countries.<sup>27</sup> The modified panel provides a broader representation of developing and emerging countries around the world and heterogeneity in terms of economic development, geographical characteristics, exchange rate regimes, etc. For this set of 26 countries, we again explore the relationship between the external public debt and the real effective exchange rate with the Empirical Iterative Bayes estimators, over the period from 1975 to 2017.

Table B.3 in the Appendix reports the coefficients get with the Bayesian shrinkage estimators for the model given by the Equation 5 which include the external public debt squared variable in the set of explanatory variables as a proxy of high or unsustainable external debt levels. We find that a low external public debt-to-GDP ratio induces a real currency appreciation, whereas a high external public debt-to-GDP ratio is associated with a depreciation of the REER. Therefore, even after including other developing countries with some heterogeneity in the panel, we find similar real exchange rate responses to external debt depending on the debt sustainability as in section 6.

Figure 2 shows the countries' public external debt between 1975 and 2017 and the

<sup>&</sup>lt;sup>27</sup>We keep Bangladesh, Benin, Botswana, Burkina Faso, Cameroon, Cote d'Ivoire, Honduras, Lesotho, Mali, Mauritania, Mauritius, Philippines, Republic of Congo, Senegal and Togo in the panel. And we add Dominican Republic, Egypt, El Salvador, Gabon, Guatemala, Jamaica, Panama, Paraguay, Sri Lanka, Thailand and Tunisia.

threshold levels for each country at which the relationship between external public debt and the RER shifts from positive to negative. The debt threshold (red dashed line) corresponds to the turning point in the relationship between external public debt and the real exchange rate. We derive this turning point from the estimated coefficients (Equation 5) reported in Table B.3 in the Appendix. The external public debt thresholds are also presented in this table for each country. Most of the time, developing countries have an external public debt-to-GDP ratio that exceeds the threshold.

Real currency appreciation occurs once the debt ratio falls below the threshold, which is the case for most developing countries. For example, for HIPCs, the external debt-to-GDP ratio drops below the threshold thanks to the debt relief granted under the HIPC and MDRI initiatives. The HIPC and MDRI debt relief period is represented by the gray area in Figure 2. The aim of the debt write-off was, on the one hand, to help these countries to recover sustainable debt levels and, on the other, to release a fiscal space to finance social expenditures and public investments. This spending may increase the demand for non-tradable goods and lead to an appreciation of the RER. At first glance, the external public debt-to-GDP ratios that minimize changes in RER may seem quite low (18 percent of GDP on average) compared to the historical external debt levels reached by some developing countries, especially HIPCs. However, they are relatively close to the external public debt levels at the end of the debt relief process that helped HIPCs to recover external debt sustainability. We will detail the implications for economic policy and external financing over the next few years in the conclusion.





currency appreciation. This threshold is derived from the estimated coefficients in Equation 5 and reported in Table ?? in the Appendix. The grey zone indicates the years of debt relief granted to HIPCs. Dominican Republic, El Salvador, Panama, Sri Lanka, and Tunisia are not shown because their minimum debt ratio never crosses the threshold. Notes: The blue line represents the countries' public external debt-to-GDP ratio from 1975 to 2017. The red dash line is the threshold for the change in the real exchange rate response to external debt. External indebtedness above the line is associated with a real currency depreciation, while under the line it is associated with a real

## 8 Conclusion

In this chapter, we seek to provide empirical evidence to inform the discussion about the real exchange rate's response to external public indebtedness in developing countries. We find an inverted U-shape relationship between the external public debt and the RER.

On the one hand, a limited external public debt-to-GDP ratio is associated with an appreciation of the RER. Capital inflows induced by external financing increase demand in the non-tradable sector and cause an appreciation of the real exchange rate. Although external financing helps governments to finance their development needs (DORNBUSCH, 1998) and offset the lack of savings, external public indebtedness leads to a real currency appreciation that results in a loss in external competitiveness and affects economic growth. Some developing countries, especially Sub-Saharan African and Latin America countries, experienced an associated increasing external debt and real currency appreciation during the 1980's and 1990's.

On the other hand, a high debt-to-GDP ratio is associated with a depreciation of the RER. Traditional approaches emphasize that external indebtedness is not sustainable for the current account over the long term. This real currency depreciation is expected to restore the external macroeconomic balance. Moreover, a high debt burden may lead the government to reduce social expenditures in order to service the debt. This results in a drop in the demand for non-tradable goods, and thus to a real currency depreciation.

The governments in developing countries must deal with the appreciation and depreciation of the RER based on the changes in external public debt-to-GDP ratio, through their economic policies. The RER is a fundamental relative price that guides investment choices between different sectors in an economy, i.e., tradable or non-tradable sectors.

In addition to the risks associated with a high debt burden, a fall in the RER may lead to an overvaluation of the currency for countries with fixed-exchange-rate regime, especially in a currency union. A depreciation of the RER can promote exports and manufacturing sectors, even though the discussions on the likely end of export-led growth strategies especially for developing countries are not vet over (RODRIK, 2016; DIAO et al., 2021; KRUSE et al., 2023). However, it may cause inflation and affect the labor productivity. Indeed, a real currency depreciation corresponds to a decrease in real income (in terms of tradable goods) for workers, and lead to inflation which affects the poor. The decline of real income may alter the health and the motivation of workers and reduce the returns of education.<sup>28</sup> It results in a decrease in labor productivity and the average level of education in the country or provoke a brain drain.<sup>29</sup> Therefore, real exchange rate stability remains an important growth lever (conditional convergence with education and skills). Growth and development opportunities are also possible through the farming sector and diversification in crop exports. Furthermore, a depreciation of the RER may also have detrimental effects for emerging market economies with domestic liability dollarization. The real depreciation makes it harder for firms in the non-tradable sector to repay their loans, and can trigger uncertainty about the banking sector solvency, or even banking panic (CALVO et al., 2008).

Therefore, there is a strong incentive for developing countries to ensure that their external public debt-to-GDP should converge towards the threshold that minimizes RER fluctuations. This threshold corresponds to the turning point in the relationship between external public debt and the RER. In the same way as studies that attempt to identify levels of external debt that do not affect economic growth, it is also possible to determine a level of external public debt to GDP that minimizes changes in the RER.

 $<sup>^{28}</sup>$ See the "X-efficiency" developed by LEIBENSTEIN (1966)

 $<sup>^{29}</sup>$  Guillaumont Jeanneney and Hua (2011) highlight transmission channels of RER on productivity in China.

We find that the ratio of external public debt-to-GDP, which stabilizes variations in the RER, is close to the debt levels at the end of HIPC initiatives, suggesting that governments have a *little room* for external indebtedness.

In 2017, HIPCs such as Benin, Cameroon and Mali reached this debt threshold, and Honduras and Senegal had already exceeded it. However, developing countries require large amounts of external financing to face economic turmoil and other challenges (climate change, digitalization, diversification), at the same time as funding costs are increasing. This means that developing countries will have to find and use financing sources that avoid returning to unsustainable debt levels, for example the use of the SDR allocation of August 2021. Or even implementing the conditions to attract private financing such as foreign direct investments or other non-debt creating flows.

Lastly, the debt threshold is not the same for all countries. The minimum debt threshold is 6.60 for Guatemala while the maximum is 35.06 for the Republic of Congo. Countries with lower debt thresholds have narrower margins for external public debt. These differences in debt thresholds could be the subject of future investigations.

## References

- ACOSTA, P. A., LARTEY, E. K. K., MANDELMAN, F. S. (2009). "Remittances and the Dutch Disease". Journal of International Economics 79.1, pp. 102–116 (cit. on p. 12).
- ARTIS, M. J., TAYLOR, M. P. (1993). "DEER Hunting: Misalignment, Debt Accumulation, and Desired Equilibrium Exchange Rates". *IMF Working Papers* 1993.048 (cit. on p. 2).
- BACHA, E. L. (1990). "A Three-Gap Model of Foreign Transfers and the GDP Growth Rate in Developing Countries". *Journal of Development Economics* 32.2, pp. 279–296 (cit. on p. 6).
- BALTAGI, B. H., BRESSON, G., PIROTTE, A. (2008). "To Pool or Not to Pool?" The Econometrics of Panel Data: Fundamentals and Recent Developments in Theory and Practice. Ed. by László MÁTYÁS and Patrick SEVESTRE. Advanced Studies in Theoretical and Applied Econometrics. Berlin, Heidelberg: Springer, pp. 517–546 (cit. on pp. 21, 22).
- BLANCHARD, O. J. (1985). "Debt, Deficits, and Finite Horizons". Journal of Political Economy 93.2, pp. 223–247 (cit. on p. 7).
- BUITER, W. H. (1988). "Death, Birth, Productivity Growth and Debt Neutrality". *The Economic Journal* 98.391, pp. 279–293 (cit. on p. 8).
- CALVO, G. A., IZQUIERDO, A., MEJÍA, L.-F. (2008). Systemic Sudden Stops: The Relevance Of Balance-Sheet Effects And Financial Integration. National Bureau of Economic Research: 14026. Pre-published (cit. on p. 28).
- CLARK, P. B., MACDONALD, R. (1999). "Exchange Rates and Economic Fundamentals: A Methodological Comparison of Beers and Feers". *Equilibrium Exchange Rates*. Ed. by Ronald MACDON-ALD and Jerome L. STEIN. Recent Economic Thought Series. Dordrecht: Springer Netherlands, pp. 285–322 (cit. on pp. 1, 5, 12).
- COMBES, J.-L., KINDA, T., PLANE, P. (2012). "Capital Flows, Exchange Rate Flexibility, and the Real Exchange Rate". *Journal of Macroeconomics* 34.4, pp. 1034–1043 (cit. on p. 12).
- COTTANI, J. A., CAVALLO, D. F., KHAN, M. S. (1990). "Real Exchange Rate Behavior and Economic Performance in LDCs". *Economic Development and Cultural Change* 39.1, pp. 61–76 (cit. on p. 1).
- COUDERT, V. (1999). "Comment Definir Un Taux de Change d'equilibre Pour Les Pays Emergents ?" Economie Internationale 77, pp. 45–66 (cit. on pp. 7, 22).
- COUHARDE, C., DELATTE, A.-L., GREKOU, C., MIGNON, V., MORVILLIER, F. (2018). "EQCHANGE: A World Database on Actual and Equilibrium Effective Exchange Rates". *International Economics* 156, pp. 206–230 (cit. on pp. 10, 11).
- COUHARDE, C., REY, S., SALLENAVE, A. (2016). "External Debt and Real Exchange Rates' Adjustment in the Euro Area: New Evidence from a Nonlinear NATREX Model". Applied Economics 48.11, pp. 966–986 (cit. on p. 8).
- DIAO, X., ELLIS, M., MCMILLAN, M. S., RODRIK, D. (2021). Africa's Manufacturing Puzzle: Evidence from Tanzanian and Ethiopian Firms. National Bureau of Economic Research: 28344. Prepublished (cit. on p. 28).
- DORNBUSCH, R. (1998). "Capital Controls: An Idea Whose Time Is Past". Should the IMF Pursue Capital-Account Convertibility? Essays in International Finance. Fischer, S., et al. (Eds.) International Monetary Fund. 207. Washington, D.C. (cit. on p. 27).
- EDWARDS, S. (1988). "Real and Monetary Determinants of Real Exchange Rate Behavior: Theory and Evidence from Developing Countries". *Journal of Development Economics* 29.3, pp. 311–341 (cit. on pp. 7, 13, 15).
- (1989). "Exchange Rate Misalignment In Developing Countries". The World Bank Research Observer 4.1, pp. 3–21 (cit. on pp. 1, 23).

- ELBADAWI, I. (1994). "Estimating Long-run Equilibrium Real Exchange Rates". Estimating Equilibrium Exchange Rates, J. Williamson. Institute for International Economics. Washington, DC (cit. on p. 13).
- FARUQEE, H. (1995). "Long-Run Determinants of the Real Exchange Rate: A Stock-Flow Perspective". Staff Papers 42.1, pp. 80–107 (cit. on pp. 2, 4, 16).
- FRENKEL, J. A., GOLDSTEIN, M. (1986). A Guide to Target Zones. National Bureau of Economic Research: 2113. Pre-published (cit. on p. 4).
- FRENKEL, J. A., RAZIN, A. (1996). Fiscal Policies and Growth in the World Economy. MIT Press. 668 pp. (cit. on p. 8).
- GANELLI, G. (2005). "The New Open Economy Macroeconomics of Government Debt". Journal of International Economics 65.1, pp. 167–184 (cit. on p. 7).
- GHURA, D., GRENNES, T. J. (1993). "The Real Exchange Rate and Macroeconomic Performance in Sub-Saharan Africa". *Journal of Development Economics* 42.1, pp. 155–174 (cit. on p. 1).
- GUILLAUMONT JEANNENEY, S., HUA, P. (2011). "How Does Real Exchange Rate Influence Labour Productivity in China?" *China Economic Review*. Sustainable Natural Resource Use in Rural China 22.4, pp. 628–645 (cit. on p. 28).
- HSIAO, C. (2022). Analysis of Panel Data. Cambridge University Press. 539 pp. (cit. on pp. 21, 39).
- HSIAO, C., PESARAN, H., TAHMISCIOGLU, A. (1998). "Bayes Estimation of Short-run Coefficients in Dynamic Panel Data Models" (cit. on pp. 21, 22, 39).
- ILZETZKI, E., REINHART, C. M., ROGOFF, K. S. (2019). "Exchange arrangements entering the twentyfirst century: Which anchor will hold?" *The Quarterly Journal of Economics* 134.2, pp. 599–646 (cit. on p. 34).
- IM, K. S., PESARAN, M. H., SHIN, Y. (2003). "Testing for Unit Roots in Heterogeneous Panels". Journal of Econometrics 115.1, pp. 53–74 (cit. on p. 17).
- KRUSE, H., MENSAH, E., SEN, K., VRIES, G. (2023). "A Manufacturing (Re)Naissance? Industrialization in the Developing World". *IMF Economic Review* 71.2, pp. 439–473 (cit. on p. 28).
- LANE, P., MILESI-FERRETTI, G. M. (2000). "The External Wealth of Nations". *Centre for Economic Policy Research Discussion Paper* 2231 (cit. on p. 16).
- LARTEY, E. K. K. (2008). Capital Inflows, Dutch Disease Effects, and Monetary Policy in a Small Open Economy (cit. on p. 2).
- LARTEY, E. K. K., MANDELMAN, F. S., ACOSTA, P. A. (2012). "Remittances, Exchange Rate Regimes and the Dutch Disease: A Panel Data Analysis". *Review of International Economics* 20.2, pp. 377– 395 (cit. on p. 15).
- LEIBENSTEIN, H. (1966). "Allocative Efficiency vs. "X-Efficiency"". *The American Economic Review* 56.3, pp. 392–415. JSTOR: 1823775 (cit. on p. 28).
- LIM, G. C., STEIN, J. L. (1998). "The Dynamics of the Real Exchange Rate and Current Account in a Small Open Economy: Australia". *Fundamental Determinants of Exchange Rates*. Ed. by Jerome L. STEIN and Polly Reynolds ALLEN. Oxford University Press, p. 0 (cit. on pp. 1, 5).
- MADDALA, G. S., HU, W. (1996). "The Pooling Problem". The Econometrics of Panel Data: A Handbook of the Theory with Applications. Ed. by László MÁTYÁS and Patrick SEVESTRE. Dordrecht: Springer Netherlands, pp. 307–322 (cit. on p. 21).
- MADDALA, G. S., TROST, R. P., LI, H., JOUTZ, F. (1997). "Estimation of Short-Run and Long-Run Elasticities of Energy Demand From Panel Data Using Shrinkage Estimators". Journal of Business & Economic Statistics 15.1, pp. 90–100 (cit. on pp. 20, 21, 39).

- MONGARDINI, J. (1998). "Estimating Egypt's Equilibrium Real Exchange Rate". *IMF Working Papers* 1998.005 (cit. on p. 7).
- NELSON, C. R., PLOSSER, C. R. (1982). "Trends and Random Walks in Macroeconmic Time Series: Some Evidence and Implications". *Journal of Monetary Economics* 10.2, pp. 139–162 (cit. on p. 22).
- OBSTFELD, M., ROGOFF, K. (1995). "Exchange Rate Dynamics Redux". Journal of Political Economy 103.3, pp. 624–660 (cit. on p. 7).
- PEDRONI, P. (2004). "Panel Cointegration: Asymptotic And Finite Sample Properties Of Pooled Time Series Tests With An Application To The PPP Hypothesis". *Econometric Theory* 20.3, pp. 597– 625 (cit. on p. 19).
- PESARAN, M. H., SHIN, Y., SMITH, R. P. (1999). "Pooled Mean Group Estimation of Dynamic Heterogeneous Panels". *Journal of the American Statistical Association* 94.446, pp. 621–634 (cit. on p. 14).
- PESARAN, M. H., SMITH, R. (1995). "Estimating Long-Run Relationships from Dynamic Heterogeneous Panels". Journal of Econometrics 68.1, pp. 79–113 (cit. on p. 14).
- PESARAN, M. (2003). A Simple Panel Unit Root Test in the Presence of Cross Section Dependence. Tech. rep. Faculty of Economics, University of Cambridge (cit. on p. 17).
- RAZIN, O., COLLINS, S. M. (1997). *Real Exchange Rate Misalignments and Growth*. National Bureau of Economic Research: 6174. Pre-published (cit. on p. 1).
- RODRIK, D. (2008). "The Real Exchange Rate and Economic Growth". Brookings Papers on Economic Activity 2008.2, pp. 365–412 (cit. on pp. 1, 10, 16).
- (2016). "Premature Deindustrialization". Journal of Economic Growth 21.1, pp. 1–33 (cit. on p. 28).
- ROGOFF, K. (1996). "The Purchasing Power Parity Puzzle". Journal of Economic Literature 34.2, pp. 647–668. JSTOR: 2729217 (cit. on pp. 16, 21).
- ROY, R., DIXON, R. (2016). "Workers' Remittances and the Dutch Disease in South Asian Countries". *Applied Economics Letters* 23.6, pp. 407–410 (cit. on p. 12).
- SEKKAT, K., VAROUDAKIS, A. (2000). "Exchange Rate Management and Manufactured Exports in Sub-Saharan Africa". *Journal of Development Economics* 61.1, pp. 237–253 (cit. on pp. 3, 7, 8).
- SENE, B., WANE, A. (2013). "Transfer Theory Re-examined In Developing Countries: Dynamic Links Between Debt Overhang And Equilibrium Real Exchange Rates". Brussels Economic Review 56.2, pp. 191–208 (cit. on p. 8).
- SMITH, A. F. M. (1973). "A General Bayesian Linear Model". Journal of the Royal Statistical Society: Series B (Methodological) 35.1, pp. 67–75 (cit. on p. 40).
- STEIN, J. L. (1998). "The Natural Real Exchange Rate of the United States Dollar, and Determinants of Capital Flows". *Fundamental Determinants of Exchange Rates*. Ed. by Jerome L. STEIN and Polly Reynolds ALLEN. Oxford University Press, p. 0 (cit. on p. 1).
- WILLIAMSON, J. (1983). The Exchange Rate System. Institute for International Economics. 110 pp. (cit. on p. 5).
- (1994). Estimating Equilibrium Exchange Rates. Peterson Institute. 608 pp. (cit. on p. 5).
- YANO, M., NUGENT, J. B. (1999). "Aid, Nontraded Goods, and the Transfer Paradox in Small Countries". *American Economic Review* 89.3, pp. 431–449 (cit. on p. 7).

## Appendix A

	Full Sample		
Bangladesh	Egypt	Mali	Sri Lanka
Benin	El Salvador	Mauritius	Thailand
Botswana	Gabon	Mauritania	Togo
Burkina Faso	The Gambia	Nepal	Tunisia
Burundi	Guatemala	Niger	
Cameroon	Guyana	Pakistan	
Central African Republic	Honduras	Panama	
Chad	Jamaica	Paraguay	
Republic of Congo	Kenya	Philippines	
Cote d'Ivoire	Lesotho	Rwanda	
Dominican Republic	Madagascar	Senegal	

 Table A.1: Country Panels

Low-Income and Lower-Middle Income Countries

Bangladesh	Chad	Lesotho	Pakistan
Benin	Republic of Congo	Madagascar	Philippines
Botswana	Cote d'Ivoire	Mali	Rwanda
Burkina Faso	The Gambia	Mauritius	Senegal
Burundi	Guyana	Mauritania	Togo
Cameroon	Honduras	Nepal	
Central African Republic	Kenya	Niger	

Sensitivity Tests Sample (Robustness Checks)

Bangladesh	Egypt	Lesotho	Senegal
Benin	El Salvador	Mali	Sri Lanka
Botswana	Gabon	Mauritania	Thailand
Burkina Faso	Guatemala	Mauritius	Togo
Cameroon	Honduras	Panama	Tunisia
Cote d'Ivoire	Lesotho	Paraguay	
Dominican Republic	Jamaica	Philippines	

Exchange Rate Arrangement	Nun	aber of	countr	ies
	1975	1990	2005	2016
No separate legal tender or currency union	12	11	11	11
Pre-announced peg or currency board arrangement	6	H	0	1
De facto peg	1	0	0	0
De facto crawling peg	1	4	ល	4
De facto crawling band that is narrower than or equal to $+/-2\%$	1	0	9	ŋ
De facto crawling band that is narrower than or equal to $+/-5\%$	1	က	1	0
De facto moving band $+/-5\%$ / Managed floating	0	0	1	1
Freely floating	0	1	0	0
Freely falling	0	2	0	0
Dual market in which parallel market data is missing	-	0	0	0

Table A.2: Fine De Facto Exchange Rate Arrangement Classification

Notes: We remove the exchange rate arrangements that do not contain any country (from less to more flexible): Pre announced horizontal band that is narrower than or equal to +/-2%; Pre announced crawling peg, de facto moving band narrower than or equal to +/-1%; Pre announced crawling band that is narrower than or equal to +/-2% or de facto horizontal band that is narrower than or equal to +/-2%; Pre announced crawling band that is wider than or equal to +/-2%; Moving band that is narrower than or equal to +/-2%. Source: ILZETZKI et al. (2019).

Countries	Decision Point	Completion Point	MDRI
Countries	Entry	Exit	
Burkina Faso	2000	2002	2005
Benin	2000	2003	2005
Senegal	2000	2004	2005
Rwanda	2000	2005	2005
Guyana	2000	2003	2005
Honduras	2000	2005	2005
Mauritania	2000	2002	2005
Mali	2000	2003	2005
Niger	2000	2004	2005
Madagascar	2000	2004	2005
The Gambia	2000	2007	2007
Cameroon	2000	2006	2006
Chad	2001	2015	2015
Burundi	2005	2009	2009
Congo, Rep.	2006	2010	2010
Central African Rep.	2007	2009	2009
Togo	2008	2010	2010
Cote d'Ivoire	2009	2012	2012

 Table A.3: Heavily Indebted Poor Countries

Source: International Monetary Fund.

Figure A.1: External public debt (in % of GDP) and Equilibrium Real Exchange Rate (in log) from 1975 to 2017



Notes: The blue line represents the country's external public debt-to-GDP, while the red line indicates movements in the equilibrium real exchange rate. Source: author's calculations; International Debt Statistics, World Development Indicators, CEPII's database "EQCHANGE".





Notes: The red line represents the fitted line from the dots. Source: author's calculations; International Debt Statistics, World Development Indicators, CEPII's database "EQCHANGE".

tests
root
Unit
A.4:
Table

	Level			First differe	nce	
	IPS	ADF	CADF	IPS	ADF	CADF
REER	$-3.219^{***}$	-3.041***	-5.512***	-29.472***	-21.767***	-16.664***
Terms of Trade	$-2.340^{***}$	-1.322	-7.795***	$-32.254^{***}$	$-24.109^{***}$	-22.007***
Trade	-1.480	-1.187	-0.043	$-35.068^{***}$	-22.827***	$-17.016^{***}$
Productivity	-4.782***	$-0.198^{***}$	0.204	$-25.178^{***}$	$-21.926^{***}$	$-16.351^{***}$
NFA	-0.7620	-0.9633	-4.313***	-27.443***	-18.348***	-15.082***
<b>External Public Debt</b>	-0.944	-0.859	-2.159**	$-30.036^{***}$	$-19.214^{***}$	$-15.307^{***}$
Debt Service	-8.080***	-6.171***	-7.389***	-43.290***	-25.703***	-21.082***

Notes: \*\* and \*\*\* denote significance at 5% and 1% levels. The null hypothesis is the presence of unit root.

### **Appendix B - Empirical Iterative Bayes Estimation**

We apply the Empirical Iterative Bayes' estimator proposed by MADDALA et al. (1997) since it provides a heterogeneous (country by country) estimation of the relationship between external public debt and real exchange rate. As MADDALA et al. (1997) argued, in panel analysis, the problem with the two usual estimation methods of either pooling the data or obtaining separate estimates for each cross-section is that both are based on extreme assumptions. If the data are pooled, it is assumed that the parameters are all the same. On the other hand, if separate estimates are obtained for each country, it is assumed that the parameters are all different. The truth probably lies somewhere in between. The parameters are not the same, but there is some similarity between them. One way of allowing for the similarity is to assume that all the parameters come from a joint distribution with a common mean and a nonzero covariance matrix.

The authors argued that the resulting parameter estimates will be a weighted average of the overall pooled estimate and the separate time series estimates based on each cross-section. Thus, each cross-section estimate is "shrunk" toward the overall pooled estimate (i.e., "shrinkage estimator"). In this way, the solution relies on the use of a random-coefficient model in which the parameters are assumed to come from a common distribution.

According to MADDALA et al. (1997), the shrinkage estimator should be preferred if the model contains lagged endogenous variables (as is the case in the dynamic models) because it gives much more reasonable parameter values than the heterogeneous estimators. Similarly, HSIAO et al. (1998) and HSIAO (2022) confirmed that, firstly, when the time-series dimension of panel sets is large, the heterogeneous estimators are more appropriate from an econometric point of view, and secondly, in the case of panel data models with coefficient heterogeneity, the Bayesian approach performs fairly well, even when the time dimension is small. Let us consider the Bayesian approach to Equation 5, which can be rewritten in the framework of the random-coefficients model, with the following specification:

$$y_i = X_i \gamma_i + u_i \tag{6}$$

where  $y_i$  contains the annual change in the ratio of external public debt to GDP time series, X is the matrix with explanatory variables, and  $\gamma_i$  are slope coefficients. In the Bayesian framework, the prior distribution of  $\gamma_i$  is given by:  $\gamma_i \sim N(\mu, \Sigma)$  where the parameters  $\mu$  (mean of  $\gamma_i$ ),  $\Sigma$  (variance of  $\gamma_i$ ) and  $\sigma_i^2$  (residual variance) are unknown. Therefore some assumptions have to be made on the prior specification of these parameters. Then we can derive the posterior distribution for the parameters,  $\gamma_i$ . On the other hand, if  $\mu, \Sigma$  and  $\sigma_i^2$  are all known, the posterior distribution of  $\gamma_i$  is normal and calculated by:

$$\gamma_i^* = \left[\frac{1}{\sigma_i^{*2}} X_i' X_i + \Sigma^{*-1}\right]^{-1} \left[\frac{1}{\sigma_i^{*2}} X_i' X_i \hat{\gamma}_i + \Sigma^{*-1} \mu^*\right]$$
(7)

where  $\hat{\gamma}_i$  is the ordinary least squares (OLS) estimator of  $\gamma_i$ . The posterior distribution of mean  $\gamma_i$  and its variance are shown in Equations 8 and 9 respectively.

$$\mu^* = \frac{1}{N} \sum_{i=1}^N \gamma_i^* \tag{8}$$

$$V[\gamma_i^*] = \left[\frac{1}{\sigma_i^{*2}} X_i' X_i + \Sigma^{*-1}\right]^{-1}$$
(9)

Since in general,  $\Sigma$  and  $\sigma_i^2$  are unknown parameters, one needs to specify priors for them. For this purpose, SMITH (1973) suggested using the mode of the joint posterior distribution given by the following equations:

$$\sigma_i^{*2} = \frac{1}{T + \varsigma_i + 2} \left[ \varsigma_i \lambda_i + \left( y_i - X_i \gamma_i^* \right)' \left( y_i - X_i \gamma_i^* \right) \right]$$
(10)

and

$$\Sigma^* = \frac{1}{T - k - 2 + \delta} \left[ R + \sum_{i=1}^N \left( \gamma_i^* - \mu^* \right) \left( \gamma_i^* - \mu^* \right)' \right]$$
(11)

where the parameters  $\varsigma_i, \lambda_i, \delta$  and R arise from the specification of the prior distributions. Moreover, SMITH (1973) proposed the approximation of these parameters by setting  $\varsigma_i = 0, \delta = 1$  and R as a diagonal matrix with small positive entries (e.g., 0.001). By doing so, the estimators take the following forms:

$$\sigma_i^{*2} = \frac{1}{T+2} \left[ (y_i - X_i \gamma_i^*)' (y_i - X_i \gamma_i^*) \right]$$
(12)

$$\Sigma^* = \frac{1}{T - k - 1} \left[ R + \sum_{i=1}^{N} \left( \gamma_i^* - \mu^* \right) \left( \gamma_i^* - \mu^* \right)' \right]$$
(13)

$$\gamma_i^* = \left[\frac{1}{\sigma_i^{*2}} X_i' X_i + \Sigma^{*-1}\right]^{-1} \left[\frac{1}{\sigma_i^{*2}} X_i' X_i \hat{\gamma}_i + \Sigma^{*-1} \mu^*\right]$$
(14)

and

$$\mu^* = \frac{1}{N} \sum_{i=1}^{N} \gamma_i^*$$
 (15)

$$V[\gamma_i^*] = \left[\frac{1}{\sigma_i^{*2}} X_i' X_i + \Sigma^{*-1}\right]^{-1}$$
(16)

Then Equations (12-16) must be solved iteratively, with the initial iteration using the OLS estimator  $\hat{\gamma}_i$  to compute  $\mu^*, \Sigma^*$  and  $\sigma_i^{*2}$ . The second and subsequent iterations are based on the empirical iterative Bayes estimator  $\gamma_i^*$ .

Country	Variable	Parameters	T-Stat	Country	Parameters	T-Stat
BGD	Const	0.6879	6.3465	KEN	1.3458	7.9578
	REER	-0.3069	-15.1709		-0.1730	-4.4413
	Debt	0.5602	4.1592		-0.2556	-1.2340
	Debt2	-0.0972	-4.4071		0.0217	0.6762
BEN	Const	1.0160	5.0206	LSO	1.0405	4.9842
	REER	-0.2392	-5.4946		-0.2251	-5.2645
	Debt	0.1544	0.6188		0.1289	0.4983
	Debt2	-0.0353	-0.9413		-0.0280	-0.7385
BWA	Const	0.9850	5.8613	MDG	0.5305	3.7341
	REER	-0.2345	-7.1780		-0.3234	-11.0388
	Debt	0.1992	0.9533		0.7633	4.3391
	Debt2	-0.0359	-1.1458		-0.1247	-4.8548
BFA	Const	1.0115	6.4672	MLI	0.9799	4.7570
	REER	-0.2320	-7.4405		-0.2369	-5.4005
	Debt	0.1635	0.8428		0.2043	0.8022
	Debt2	-0.0389	-1.2913		-0.0380	-1.0503
BDI	Const	0.9546	9.7399	MRT	1.1850	7.3383
	REER	-0.2437	-12.7794		-0.1892	-5.4334
	Debt	0.2355	1.9306		-0.0473	-0.2369
	Debt2	-0.0409	-2.2877		-0.0009	-0.0338
CMR	Const	1.0824	6.2020	MUS	0.9744	13.3274
	REER	-0.2239	-5.6084		-0.2470	-20.1404
	Debt	0.0733	0.3420		0.2052	2.2458
	Debt2	-0.0213	-0.6909		-0.0463	-2.8593
CAF	Const	0.6064	2.9802	NPL	1.0113	26.6401
	REER	-0.3170	-7.1179		-0.2458	-43.2371
	Debt	0.6650	2.6517		0.1575	3.3200
	Debt2	-0.1097	-3.0134		-0.0357	-4.1307
TCD	Const	1.0820	5.2194	NER	1.3371	8.6642
	REER	-0.2148	-4.9795		-0.1584	-5.2840
	Debt	0.0771	0.3009		-0.2366	-1.2338
	Debt2	-0.0277	-0.7128		0.0219	0.7519
COG	Const	1.1226	10.5553	$\mathbf{PAK}$	1.4948	8.0949
	REER	-0.2130	-8.1949		-0.1304	-3.3473
	$\operatorname{Debt}$	0.0253	0.1945		-0.4336	-1.9011
	Debt2	-0.0116	-0.6658		0.0543	1.5814
CIV	Const	1.0498	7.1855	$_{\rm PHL}$	0.7548	3.9420
	REER	-0.2284	-6.7980		-0.2998	-7.2803
	$\operatorname{Debt}$	0.1150	0.6408		0.4745	2.0096
	Debt2	-0.0261	-1.0257		-0.0824	-2.3000
GMB	Const	1.0573	7.8601	RWA	1.0616	10.3171
	REER	-0.2242	-7.6722		-0.2296	-10.9981
	$\operatorname{Debt}$	0.1081	0.6511		0.0986	0.7730
	Debt2	-0.0188	-0.8093		-0.0228	-1.1836
GUY	Const	1.0625	10.9006	SEN	1.0714	5.2276
	REER	-0.2238	-9.3180		-0.2204	-5.1559
	$\operatorname{Debt}$	0.1012	0.8472		0.0896	0.3535
	Debt2	-0.0199	-1.3189		-0.0242	-0.6354
HND	Const	0.7894	3.6278	TGO	1.0238	6.0151
	REER	-0.2820	-5.9546		-0.2302	-6.1475
	Debt	0.4365	1.6275		0.1488	0.7092
	Debt2	-0.0790	-1.9766		-0.0306	-1.0227

Table B.1: Shrinkage estimators country by country from Eq.5 – 1975-1999

Notes: Const, REER, Debt, Debt2 correspond to the constant, the lagged real effective exchange rate, the external public debt-to-GDP ratio, and squared variable of Debt, respectively. The number of iterations is 10.

Country	Variable	Parameters	T-Stat	Country	Parameters	T-Stat
BGD	Const	0.3970	63.0947	KEN	0.3979	63.2196
	REER	-0.1117	-14.1645		-0.0967	-11.4798
	Debt	0.0902	7.0061		0.0768	5.8770
	Debt2	-0.0164	-5.5121		-0.0196	-7.5582
BEN	Const	0.3969	63.3247	LSO	0.3980	63.0504
	REER	-0.1084	-14.7415		-0.0969	-10.8979
	Debt	0.0856	6.6278		0.0782	5.9492
	Debt2	-0.0165	-6.7820		-0.0170	-6.2051
BWA	Const	0.3976	63.1772	MDG	0.3983	63.2900
	REER	-0.0991	-12.4037		-0.0977	-11.8008
	Debt	0.0789	6.0699		0.0809	6.1693
	Debt2	-0.0163	-3.9223		-0.0175	-7.2687
BFA	Const	0.3980	63.2896	MLI	0.3970	63.0641
	REER	-0.0979	-12.3582		-0.1048	-12.2407
	$\operatorname{Debt}$	0.0790	6.0997		0.0828	6.2723
	Debt2	-0.0187	-6.4056		-0.0147	-6.4494
BDI	Const	0.3983	63.5820	MRT	0.3976	63.0832
	REER	-0.0961	-13.9912		-0.0923	-10.4302
	$\operatorname{Debt}$	0.0796	6.2465		0.0721	5.5045
	Debt2	-0.0157	-7.3977		-0.0145	-6.8573
CMR	Const	0.3972	63.5381	MUS	0.3974	63.7264
	REER	-0.1054	-15.3927		-0.1020	-17.4057
	$\operatorname{Debt}$	0.0845	6.5519		0.0805	6.4854
	Debt2	-0.0158	-7.0163		-0.0185	-6.9827
CAF	Const	0.3986	63.4275	NPL	0.3976	63.7112
	REER	-0.0958	-12.0752		-0.1043	-20.7800
	$\operatorname{Debt}$	0.0803	6.1259		0.0853	7.2289
	Debt2	-0.0172	-7.2307		-0.0180	-7.6533
TCD	Const	0.3981	63.2522	NER	0.3982	63.3821
	REER	-0.0981	-12.2116		-0.0915	-12.3278
	$\operatorname{Debt}$	0.0794	6.0919		0.0731	5.7253
	Debt2	-0.0186	-6.8142		-0.0181	-6.9542
COG	Const	0.3967	63.0945	$\mathbf{PAK}$	0.3973	63.0629
	REER	-0.1084	-12.5078		-0.0975	-10.8139
	$\operatorname{Debt}$	0.0854	6.4042		0.0759	5.7274
	Debt2	-0.0138	-7.0196		-0.0161	-5.7555
CIV	Const	0.3974	63.1946	$_{\rm PHL}$	0.3975	63.3350
	REER	-0.1041	-12.0037		-0.1038	-13.2333
	Debt	0.0844	6.3469		0.0836	6.4133
	Debt2	-0.0153	-6.9750		-0.0176	-6.7905
GMB	Const	0.3979	63.1277	RWA	0.3983	63.7141
	REER	-0.0952	-10.8498		-0.0943	-13.9074
	Debt	0.0766	5.8130		0.0753	5.8844
	Debt2	-0.0153	-6.2750	~~~~	-0.0178	-7.4694
GUY	Const	0.3969	63.1387	SEN	0.3980	63.2372
	REER	-0.1080	-12.6140		-0.0972	-11.5870
	Debt	0.0853	6.3860		0.0781	5.9504
UND	Debt2	-0.0131	-7.2859	maa	-0.0170	-6.5296
HND	Const	0.3980	63.1177	TGO	0.3972	63.1672
	REER	-0.0996	-11.1349		-0.1046	-12.2306
	Debt	0.0814	6.1370		0.0838	6.3374
	Debt2	-0.0179	-6.9326		-0.0147	-6.7028

Table B.2: Shrinkage estimators country by country from Eq.5 – 1975-2017

Notes: Const, REER, Debt, Debt2 correspond to the constant, the lagged real effective exchange rate, the external public debt-to-GDP ratio, and squared variable of Debt, respectively. The number of iterations is 12.

Country	Variable	Parameters	T-Stat	Threshold	Country	Parameters	T-Stat	Threshold
BGD	Const	0.6078	3.9081		$_{ m JAM}$	0.4870	2.5321	
	REER	-0.1905	-6.3610			-0.1587	-3.9398	
	$\operatorname{Debt}$	0.2079	13.8078	17		0.2035	12.7676	16.82
	$\mathrm{Debt2}$	-0.0367	-12.7265			-0.0361	-14.9294	
BEN	$\operatorname{Const}$	0.5107	2.7277		$\mathrm{LSO}$	0.1602	0.9483	
	REER	-0.1739	-4.4363			-0.0881	-2.5610	
	Debt	0.2092	14.4373	18.36		0.1989	12.6462	17.21
	$\mathrm{Debt2}$	-0.0359	-14.5290			-0.0350	-13.1213	
BWA	$\operatorname{Const}$	0.0593	0.3349		MLI	0.1950	1.3126	
	REER	-0.0652	-1.8118			-0.1112	-3.5679	
	$\operatorname{Debt}$	0.1950	12.3103	14.77		0.2124	13.6713	27.26
	$\mathrm{Debt2}$	-0.0362	-8.6787			-0.0321	-14.3450	
BFA	$\operatorname{Const}$	0.2670	1.5237		$\operatorname{MRT}$	-0.1108	-0.9944	
	REER	-0.1075	-3.0711			-0.0375	-1.6291	
	$\operatorname{Debt}$	0.1956	12.8835	13.58		0.2028	12.9684	24.18
	$\mathrm{Debt2}$	-0.0375	-12.9819			-0.0318	-15.3325	
CMR	$\operatorname{Const}$	0.4251	2.2450		MUS	0.2814	1.6400	
	REER	-0.1545	-3.7777			-0.1102	-2.9414	
	$\mathbf{Debt}$	0.2093	14.5637	21.32		0.1929	14.2415	12.38
	$\mathrm{Debt2}$	-0.0342	-15.1113			-0.0383	-14.7374	
COG	$\operatorname{Const}$	0.4031	1.9510		PAN	-0.0370	-0.3862	
	REER	-0.1642	-3.6765			-0.0419	-2.0410	
	$\mathbf{Debt}$	0.2199	14.1213	35.06		0.1941	12.3391	15.45
	Debt2	-0.0309	-15.1800			-0.0355	-14.8663	
CIV	$\operatorname{Const}$	0.3762	1.8693		PRY	0.3122	1.6365	
	REER	-0.1494	-3.3764			-0.1129	-2.7172	
	Debt	0.2142	13.7799	25.72		0.1913	11.5684	11.54
						0	ontinued o	n next page

Table B.3: Sensitivity Robustness Checks - Shrinkage estimators country by country from Eq.5 – 1975-2017

Country	Variable	Parameters	<b>T-Stat</b>	Threshold	Country	Parameters	<b>T-Stat</b>	Threshold
	Debt2	-0.0330	-15.0897			-0.0391	-10.7981	
DOM	$\operatorname{Const}$	0.5003	2.4520		PHL	0.4541	2.2046	
	REER	-0.1553	-3.6068			-0.1573	-3.5915	
	$\operatorname{Debt}$	0.1946	11.8466	11.11		0.2057	13.8479	16.34
	Debt2	-0.0404	-11.8148			-0.0368	-14.1167	
EGY	$\operatorname{Const}$	0.3053	1.5160		SEN	0.1737	0.9925	
	REER	-0.1298	-2.9530			-0.0911	-2.4938	
	Debt	0.2084	12.6586	21.95		0.1985	12.8981	16.81
	Debt2	-0.0337	-11.1314			-0.0352	-13.8326	
SLV	$\operatorname{Const}$	0.2739	1.6396		LKA	0.6497	3.3875	
	REER	-0.1220	-3.1988			-0.2010	-5.2091	
	Debt	0.2067	12.6097	18.03		0.2077	13.0853	15.41
	Debt2	-0.0357	-12.4228			-0.0380	-13.6361	
GAB	$\operatorname{Const}$	0.1463	0.8653		THA	0.3177	1.9093	
	REER	-0.0917	-2.5699			-0.1071	-2.9480	
	Debt	0.2053	12.9872	21.02		0.1811	14.5317	8.51
	Debt2	-0.0337	-13.3815			-0.0423	-14.8707	
GTM	$\operatorname{Const}$	0.3778	2.1573		TGO	0.2597	1.3403	
	REER	-0.1112	-3.0997			-0.1226	-2.8713	
	Debt	0.1744	11.8254	6.6		0.2117	13.7604	26.47
	Debt2	-0.0460	-13.0303			-0.0323	-14.8837	
HND	$\operatorname{Const}$	0.4401	2.2917		TUN	-0.0514	-0.3541	
	REER	-0.1501	-3.7452			-0.0414	-1.3821	
	$\operatorname{Debt}$	0.2031	12.9499	15.89		0.1956	12.4172	16.12
	Debt2	-0.0367	-14.0047			-0.0352	-14.5477	

Table B.3: (continued)

Notes: Const, REER, Debt, Debt2 correspond to the constant, the lagged real effective exchange rate, the external public debt-to-GDP ratio, and squared variable of Debt, respectively. The number of iterations is 8.

## DOCUMENTS DE TRAVAIL GREDEG PARUS EN 2025 GREDEG Working Papers Released in 2025

2025-01	Bruno Deffains & Frédéric Marty
	Generative Artificial Intelligence and Revolution of Market for Legal Services
2025-02	Annie L. Cot & Muriel Dal Pont Legrand
	"Making war to war" or How to Train Elites about European Economic Ideas: Keynes's Articles
	Published in L'Europe Nouvelle during the Interwar Period
2025-03	Thierry Kirat & Frédéric Marty
	Political Capitalism and Constitutional Doctrine. Originalism in the U.S. Federal Courts
2025-04	Laurent Bailly, Rania Belgaied, Thomas Jobert & Benjamin Montmartin
	The Socioeconomic Determinants of Pandemics: A Spatial Methodological Approach with
	Evidence from COVID-19 in Nice, France
2025-05	Samuel De La Cruz Solal
	Co-design of Behavioural Public Policies: Epistemic Promises and Challenges
2025-06	Jérôme Ballet, Damien Bazin, Frédéric Thomas & François-Régis Mahieu
	Social Justice: The Missing Link in Sustainable Development
2025-07	Jérôme Ballet & Damien Bazin
	Revoir notre manière d'habiter le monde. Pour un croisement de trois mouvements de pensée:
	capabilités, services écosystémiques, communs
2025-08	Frédéric Marty
	Application des DMA et DSA en France : analyse de l'architecture institutionnelle
	et des dynamiques de régulation
2025-09	Éléonore Dodivers & Ismaël Rafaï
	Uncovering the Fairness of AI: Exploring Focal Point, Inequality Aversion, and Altruism in
	ChatGPT's Dictator Game Decisions
2025-10	Hayk Sargsyan, Aleksandr Grigoryan & Olivier Bruno
	Deposits Market Exclusion and the Emergence of Premium Banks
2025-11	Lubica Stiblarova & Anna Tykhonenko
	Talent vs. Hard Work: On the Heterogeneous Role of Human Capital in FDI Across
	EU Member States
2025-12	Mathieu Chevrier
	Social Reputation as one of the Key Driver of AI Over-Reliance: An Experimental Test with
	ChatGPT-3.5
2025-13	RANIA BELGAIED
	L'accessibilité aux médecins généralistes libéraux pour la ville de Nice
2025-14	Lorenzo Corno, Giovanni Dosi & Luigi Marengo
	Behaviours and Learning in Complex Evolving Economies
2025-15	GREGORY DONNAT
	Real Exchange Rate and External Public Debt in Emerging and Developing Countries